Author’s response to reviews

Title: Interprofessional Education in Graduate Medical Education: Survey Study of Residency Program Directors

Authors:

Morhaf Al Achkar (alachkar@iupui.edu)
Matthew Hanauer (mhanauer@umail.iu.edu)
Chantel colavecchia (ccolavec@iu.edu)
Dean Seehusen (DSEEHUSEN@msn.com)

Version: 2 Date: 21 Nov 2017

Author’s response to reviews:

Dear editor,

We would like to thank you very much for considering our paper acceptable to publication. We appreciate all the helpful feedback that made our paper stronger. Below please find an item-by-item responses for the items that needed further consideration.

Best regards,

The authors

1. The abstract should contain the following sections - Background, Methods, Results and Conclusions. Please remove the 'Purpose' section and replace with with a 'Background' section.

Response: we removed the word purpose and replaced it background

2. Please include a list of 3-10 keywords after the abstract

Response: we included a list of 5 words.

3. The 'Consent to publish' refers to the consent of the participants to publish their data, in instances where individuals can be recognised from data published (i.e. specific case reports).
For this manuscript, under 'Consent to publish' please write 'not applicable' (since individuals cannot be identified from the data submitted)

Response: we replaced that with “not applicable”

4. When listing Author's contributions please list individual author's by initials.
Response: we replaced the author’s names with the initials

5. Availability of Data - who should readers make their requests out to, when requesting data?
Response: we included the main author’s name and email address

6. Submit the next version with track changes.
Response: we used track changes.